Author's response to reviews

Title: Association between Television Viewing and the Risk of Metabolic Syndrome in a Community-based Population

Authors:

Pei-Chia Chang (lisa319.tw@yahoo.com.tw)
Tsai-Chung Li (tcli@mail.cmu.edu.tw)
Ming-Tsang Wu (mingtsangwu@yahoo.com)
Chiu-Shong Liu (liucs@ms14.hinet.net)
Chia-Ing Li (a6446@www.cmuh.org.tw)
Ching-Chu Chen (chingchu@ms15.hinet.net)
Wen-Yuan Lin (wylin@mail.cmu.edu.tw)
Shin-Yuh Yang (yz123kimo@yahoo.com.tw)
Cheng-Chieh Lin (cmuh404@yahoo.com.tw)

Version: 4 Date: 21 April 2008

Author's response to reviews: see over
Dear Editor:

Thank you very much for giving us the chance to reply the reviewer’s comments for our paper "Association between Television Viewing and the Risk of Metabolic Syndrome in a Community-based Population ".

Concerning the comment of the reviewer, our reply was made as following:
21 April, 2008

Dear reviewer,

The authors would like to thank you for judicious reading of the manuscript and valuable comments which greatly improved this manuscript. We have made revision according to your comment.

Yours sincerely

Cheng-Chieh Lin
Professor,

Department of Family Medicine & School and Graduate Institute of Health Care Administration, China Medical University

China Medical University Hospital No 2, Yu-Der Road, 404, Taichung, Taiwan, R.O.C.,
1. PAGE 15: LAST PARAGRAPH. IT IS STATED THAT 'DUE TO A STRONG CORRELATION BETWEEN PHYSICAL ACTIVITY AND PHYSICAL INACTIVITY (OR SEDENTARY BEHAVIOUR), THE EFFECT OF LENGTH OF TIME SPENT WATCHING TV ON THE RISK OF HAVING METABOLIC SYNDROME COULD POSSIBLY BE CONFOUNDED BY PHYSICAL ACTIVITY' WILL NEED TO BE REVISED ACCORDINGLY.

Reply: We have corrected the sentence according to the reviewer’s suggestion.